Author's response to reviews

Title: Guidelines disobeyed - A severe complication associated with manual therapy of the cervical spine in a patient with Forestier's disease: a case report

Authors:

Maximilian J Hartel (mh@maxhartel.de)
Ulrich Seidel (ulrich.seidel@insel.ch)
Lukas Iselin (iselinlukas@mac.com)
Aristomenis K Exadaktylos (exadaktylos@exadaktylos.ch)
Lorin M Benneker (lorinmichael.benneker@insel.ch)

Version: 6 Date: 2 October 2011

Author's response to reviews: see over
Dear editors,

In this revision we have addressed the Editorial Office’s request: The subheading “Course” has been erased.

Sincerely yours

[Signature]

Maximilian J. Hartel
Corresponding Author